Author’s response to reviews

Title: Expression of NDRG2 is down-regulated in high-risk adenomas and colorectal carcinoma

Authors:

Anders Lorentzen (anbl@ruc.dk)
Lotte K Vogel (vogel@imbg.ku.dk)
Rikke H Lewinsky (lewinsky@ruc.dk)
Mona Saebo (mona.sabo@hit.no)
Camilla F Skjelbred (camilla-furu.skjelbred@sthf.no)
Sine Godiksen (sinego@imbg.ku.dk)
Geir Hoff (geir.hoff@sthf.no)
Kjell M Tveit (kjell.tveit@uus.no)
Inger Marie Bowitz Lothe (ingermariebowitzlothe@ulleval.no)
Tone Ikdahl (tone.ikdahl@uus.no)
Elin H Kure (elur@uus.no)
Cathy Mitchelmore (mitch@ruc.dk)

Version: 3 Date: 10 October 2007

Author’s response to reviews:

Formatting changes requested

References - The reference list contains all authors’ names, if there are 30 or below. The term 'et al.' is not be used.

Figure Legends -> Figure legends

Figure cropping - The figures are cropped and in the right format.